\begin{thebibliography}{10}

\bibitem{adar00free}
Eytan Adar and Bernardo~A. Huberman.
\newblock Free riding on gnutella.
\newblock {\em First Monday}, September 2000.

\bibitem{BAJ99}
Albert-La\'azl\'o, R\'eka Albert, and Hawoong Jeong.
\newblock Scale-free characteristics of random netwoks: The topology of the
  world wide web.
\newblock {\em submitted to {E}lsevier {Preprint}}, August 1999.

\bibitem{beth94valuation}
T.~Beth, M.~Borcherding, and B.~Klein.
\newblock Valuation of trust in open networks.
\newblock {\em Lecture Notes in Computer Science}, 875:3--??, 1994.

\bibitem{Bla96}
Matt Blaze.
\newblock Oblivious key escrow.
\newblock In {\em Proc. Workshop on Information Hiding}, number 1174 in LNCS,
  pages 334--343. Springer-Verlag, 1996.

\bibitem{BFL96}
Matt Blaze, Joan Feigenbaum, and Jack Lacy.
\newblock Decentralized trust management.
\newblock In {\em Proc. 17th Symposium on Security and Privacy}, pages
  164--173, Los Alamitos, 1996. IEEE Computer Society Press.

\bibitem{Bra97}
Marc Branchaud.
\newblock A survey of public key infrastructures.
\newblock Master's thesis, McGill University, Dept. of Computer Science, March
  1997.

\bibitem{CL99}
Miguel Castro and Barbara Liskov.
\newblock Practical byzantine fault tolerance.
\newblock In {\em Proc. Third Symp. on Operating Systems Design and
  Implementation}, New Orleans, February 1999.

\bibitem{chaum89untraceable}
D.~Chaum, A.~Fiat, and M.~Naor.
\newblock Untraceable electronic cash (extended abstract), 1989.

\bibitem{crary99typed}
Karl Crary, David Walker, and Greg Morrisett.
\newblock Typed memory management in a calculus of capabilities.
\newblock In {\em Conference Record of {POPL} 99: The 26th {ACM}
  {SIGPLAN}-{SIGACT} Symposium on Principles of Programming Languages, San
  Antonio, Texas}, pages 262--275, New York, NY, 1999.

\bibitem{chord:hotos}
Frank Dabek, Emma Brunskill, M.~Frans Kaashoek, David Karger, Robert Morris,
  Ion Stoica, and Hari Balakrishnan.
\newblock Building peer-to-peer systems with {C}hord, a distributed lookup
  service.
\newblock In {\em Proceedings of the 8th {W}orkshop on {H}ot {T}opics in
  {O}perating {S}ystems ({HotOS-VIII})}, Schloss Elmau, Germany, May 2001.
  {IEEE} {C}omputer {S}ociety.

\bibitem{dingledine00free}
Roger Dingledine, Michael~J. Freedman, and David Molnar.
\newblock The free haven project: Distributed anonymous storage service.
\newblock In {\em Workshop on Design Issues in Anonymity and Unobservability},
  number 2009 in LNCS, pages 67--95, 2000.

\bibitem{dingledine-cascrep}
Roger Dingledine and Paul Syverson.
\newblock Reliable {MIX} cascade networks through reputation.
\newblock In {\em Proc. Financial Cryptography}, March 2002.

\bibitem{douceur-sybil}
John~R. Douceur.
\newblock The sybil attack.
\newblock In {\em Proc. 1st International Workshop on Peer-to-Peer Systems},
  March 2002.

\bibitem{dwork-spam}
Cynthia Dwork and Moni Naor.
\newblock Pricing via processing or combating junk mail.
\newblock In Ernest~F. Brickell, editor, {\em Advances in Cryptology -- CRYPTO
  '92}, number 740 in LNCS, pages 139--147. Springer-Verlag, 1992.

\bibitem{flake02self-organization}
Gary~William Flake, Steve Lawrence, C.~Lee Giles, and Frans Coetzee.
\newblock Self-organization of the web and identification of communities.
\newblock {\em IEEE Computer}, 35(3):66--71, 2002.

\bibitem{gay01language}
David Gay and Alexander Aiken.
\newblock Language support for regions.
\newblock In {\em {SIGPLAN} Conference on Programming Language Design and
  Implementation}, pages 70--80, 2001.

\bibitem{kleinberg-smallworld}
Jon Kleinberg.
\newblock The small-world phenomenon: An algorithmic perspective.
\newblock In {\em Proc. 32nd ACM Symp. on Theory of Computing}, 2000.

\bibitem{kleinberg99clever}
Jon~M. Kleinberg.
\newblock Authoritative sources in a hyperlinked environment.
\newblock {\em Journal of the {ACM}}, 46(5):604--632, 1999.

\bibitem{LSP82}
Leslie Lamport, Robert Shostak, and Marshall Pease.
\newblock The byzantine generals problem.
\newblock {\em {ACM} Transactions on Programming Languages and Systems},
  4(3):382--401, July 1982.

\bibitem{LA98}
Raph Levien and Alexander Aiken.
\newblock Attack resistant trust metrics for public key certification.
\newblock In {\em 7th {USENIX} Security Symposium}, San Antonio, Texas, January
  1998.

\bibitem{Mau96}
Ueli Maurer.
\newblock Modelling a public-key infrastructure.
\newblock In E.~Bertino, H.~Kurth, G.~Martella, and E.~Montolivo, editors, {\em
  Computer Security -- {ESORICS} '96}, number 1146 in LNCS. Springer Verlag,
  1996.

\bibitem{maymounkov-kademlia}
Petar Maymounkov and David Mazi\`eres.
\newblock Kademlia: A peer-to-peer information system based on the {XOR}
  metric.
\newblock In {\em Proc. 1st International Workshop on Peer-to-Peer Systems},
  March 2002.

\bibitem{agorics-incentive}
Mark~S. Miller and K.~Eric Drexler.
\newblock Incentive engineering: for computational resource management.
\newblock In Bernardo Huberman, editor, {\em The Ecology of Computation}, pages
  231--266. Elsevier Science Publishers/North-Holland, 1988.

\bibitem{agorics-markets}
Mark~S. Miller and K.~Eric Drexler.
\newblock Markets and computation: Agoric open systems.
\newblock In Bernardo Huberman, editor, {\em The Ecology of Computation}, pages
  133--176. Elsevier Science Publishers/North-Holland, 1988.

\bibitem{moreton-trust}
Tim Moreton and Andrew Twigg.
\newblock Trading in trust, tokens and stamps.
\newblock In {\em Proc. Workshop on Economics of Peer-to-Peer Systems}, June
  2003.

\bibitem{P2P}
Andy Oram, editor.
\newblock {\em Peer to Peer}.
\newblock O'Reilly \& Associates, 2001.

\bibitem{page98pagerank}
Lawrence Page, Sergey Brin, Rajeev Motwani, and Terry Winograd.
\newblock The {P}age{R}ank citation ranking: Bringing order to the web.
\newblock Technical report, Stanford University, 1998.

\bibitem{Pos99}
David Post, 1999.

\bibitem{operatingthetan}
ptsc (pseudonym).
\newblock The {C}hurch of {S}cientology's supremacy over the search term
  ``{S}cientology'' on {G}oogle, February 2002.

\bibitem{RS97a}
Michael Reiter and Stuart Stubblebine.
\newblock Path independence for authentication in large-scale systems.
\newblock In {\em Proceedings of the 4th {ACM} Conference on Computer and
  Communications Security}, 1997.

\bibitem{RS97b}
Michael Reiter and Stuart Stubblebine.
\newblock Toward acceptable metrics of authentication.
\newblock In {\em Proceedings of the 1997 {IEEE} Symposium on Security and
  Privacy}, 1997.

\bibitem{Ross67}
D.~T. Ross.
\newblock The {AED} free storage package.
\newblock {\em Communications of the ACM}, 10(8):481--492, August 1967.

\bibitem{chord:sigcomm01}
Ion Stoica, Robert Morris, David Karger, M.~Frans Kaashoek, and Hari
  Balakrishnan.
\newblock Chord: A scalable peer-to-peer lookup service for internet
  applications.
\newblock In {\em Proceedings of the {ACM} {SIGCOMM} '01 Conference}, San
  Diego, California, August 2001.

\bibitem{tarah92associating}
Anas Tarah and Christian Huitema.
\newblock Associating metrics to certification paths.
\newblock In {\em European Symposium on Research in Computer Security
  ({ESORICS})}, pages 175--192, 1992.

\bibitem{tofte97regionbased}
Mads Tofte and Jean-Pierre Talpin.
\newblock Region-based memory management.
\newblock {\em Information and Computation}, 1997.

\bibitem{zooko-mn}
Bryce Wilcox-O'Hearn.
\newblock Experiences deploying a large-scale emergent network.
\newblock In {\em Proc. 1st International Workshop on Peer-to-Peer Systems},
  March 2002.

\end{thebibliography}

\chapter{Trust Metrics}
\label{tmetrics}

In this chapter, we review the literature of trust metrics, present a
quantitative framework for analyzing the attack resistance of trust
metrics. Finally, we prove tight upper and lower bounds on the attack
resistance of trust metrics given the usual scalar assumptions.

\section{The simplest trust metric}

In this section, we present the simplest possible trust metric, which
forms the basis of other trust metrics in the literature.

There are three inputs to this trust metric: a directed graph, a
designated ``seed'' node indicating the root of trust, and a
``target'' node. We wish to determine whether the target node is
trustworthy.

Each edge from $s$ to $t$ in the graph indicates that $s$ believes
that $t$ is trustworthy. The simplest possible trust metric evaluates
whether $t$ is reachable from $s$. If not, there is no reason to
believe that $t$ is trustworthy, given the data available.

In a cryptographic implementation, each node corresponds to a public
key, and each edge from $s$ to $t$ corresponds to a digitally signed
\emph{certificate}. In the usual terminology, $s$ is the
\emph{issuer}, $t$ is the \emph{subject}. The certificate itself is
some string identifying $t$, along with a digital signature of this
string generated by $s$.

The simplest trust metric is also the weakest with respect to
attacks. If an attacker is able to generate an edge from any node
reachable from the seed to a node under his control, then he can cause
arbitrary nodes to be accepted.  As the size of the reachable subgraph
increases, the risk of any such attack increases as well. Thus, the
primary focus in the literature is to present stronger trust metrics
that (hopefully) resist attacks better, while still accepting most
nodes that deserve trust.

Section~\ref{tmetric-survey}, discusses a sampling of trust metrics
previously published. Section~\ref{tmetric-quant}, presents a
quantitative notion of ``attack resistance.'' 

\section{Survey of the literature}
\label{tmetric-survey}

All trust metrics include the three inputs described above: the trust
graph, the seed node, or \emph{trust root}, and the target. Some trust
metrics add more detail to these inputs. Trust edges may contain some
conditions or restrictions, for example that $t$ is himself
trustworthy, but cannot be trusted to vouch for other nodes. In
addition, the target may be a richer assertion than merely whether a
certain node is trustworthy. For example, edges may additionally
identify a maximum dollar value, and the target may be an assertion
that the target node can be trusted with a transaction of some dollar
value.

Finally, there may be additional policies to be enforced by the trust
metric, or parameters supplied. Often, these parameters control the
overall strictness or tolerance of the trust metric.

The most general form of ``trust management'' system is represented by
PolicyMaker\cite{BFL96}, as certificates and policies can represent
arbitrary computations in Turing-complete language. Applications of
PolicyMaker tend to focus on the language of assertions rather than
trust computations over the graph, but the fully general nature of the
system allows the the latter to be implemented.

Most trust metrics in the literature assume monotonicity. More
precisely, if a monotonic trust metric accepts a node $t$ as
trustworthy when given a graph $G$, it will also accept node $t$ when
given a graph $G'$ that contains all trust edges in $G$. A primary
motivation for this assumption is scaling. In particular, it allows
for highly efficient overall distributed architectures in which all
participants need only access a small, local subset of the global
trust graph. Further, under such an assumption, there is no special
vulnerability to denial of service attacks. With the monotonicity
assumption, if an attacker can withold trust edges (i.e. cause the
verifier to evaluate a subset of the global trust graph), then it
cannot cause nodes to be accepted other than those which would be
accepted given the entire trust graph.

The simplest trust metric is clearly monotonic. Adding edges to the
trust graph can only cause more nodes to be reachable.

Another relatively simple trust metric is similar to the simplest one,
with the additional constraint that path lengths are bounded by some
parameter $k$. Thus, all nodes within a distance of $k$ edges from the
seed are accepted. A variation of this trust metric is used in X.509
systems.

Perhaps the earliest published system resembling a modern trust metric
is the Beth, Borcherding, and Klein\cite{beth94valuation}. This system
contains a set of elaborate inference rules for deriving a
``trustworthiness'' value between 0 and 1, using both ``direct trust''
and ``recommendation trust'' relationships encoded on edges. However,
further work by Reiter and Stubblebine\cite{RS97b} showed this trust
metric to be no more secure than the simple reachability criterion
described above. Note that the BBK trust metric is not monotonic.

A still earlier system is described by Tarah and
Huitema\cite{tarah92associating}, who suggest evaluating both
certificates and ``certificate paths.'' They suggest several possible
simple trust metrics, including the length of the certification path
and the minimum of involved trust values along the path. However,
because it evaluates only paths, rather than general graphs, this
system does not quite meet the definition of ``trust metric''.

Reiter and Stubblebine\cite{RS97a} presented the first trust metric
with the ability to resist nontrivial attacks. Briefly, this trust
metric counts the number of node-independent paths of bounded length
from the seed to the target. Thus, both the path length and the
threshold for the minimum number of such paths are tunable parameters
of the metric. A simple characterization of its attack resistance
follows easily: an attacker must add ``bad'' edges to nodes under his
own control to at least as many nodes as the threshold for the number
of independent paths. Even though evaluation of this trust metric is
an NP-complete problem, the experimental PathServer implementation
seems to perform reasonably well in practice.

Maurer presented an intriguing trust metric based on randomized
experiments\cite{Mau96}. Very briefly, each edge in the network has
an associated probability. The result of the trust metric is the
probability that the target is reachable from the seed in a subgraph
of the original graph where each node is present with the probability
given in the original graph.

Both implementation and analysis of the Maurer trust metric are
challenging. In addition, we present an analysis below
(Section~\ref{maurer-analysis}) suggesting that the Maurer metric is
vulnerable to an easily-mounted attack. To do so, we will first need
to prepare a framework for quantitatively evaluating the attack
resistance of a trust metric.

\section{Framework for analysis}
\label{tmetric-quant}

What does it mean for a trust metric to be attack resistant? In
this section, we present a quantitative framework for this question.

In any given attack, we assume that the attacker can add or delete
edges from the legitimate part of the trust graph, pointing to
arbitrary nodes. These edges may point directly to nodes under the
attacker's control, or perhaps to other good nodes, in order to
fool the trust metric.

We assign a \emph{cost} to each such attack. A typical cost metric is
to count the number of edges added. Assuming an attack of a given
cost, what is the highest number of bad assertions the attacker can
force to be accepted? If this number is limited, the trust metric is
attack resistant. If it can grow to the same order as the number of
good assertions even for a fairly low cost attack, then the trust
metric suffers from catastrophic failure and is not attack resistant.

\subsection{Cost metrics: node vs edge attacks}

We consider two cost metrics. The simplest is to count the number of
edges added. Another useful metric is to count the number of
``attacked '' nodes with added outedges, and to assume that any such
node may have an arbitrary number of edges added. As we will see
later, the attack-resistance properties of different trust metrics
behave differently under these two cost assumptions. Note that, for
any given attack, the number of nodes counted is no greater than the
number of edges counted.

An edge attack corresponds to fooling a victim into generating an
edge. In many cases, mounting such an attack is straightforward. For
example, the attacker may send a forged email pretending to be a
friend of the victim, asking for a cert. Many such electronic means
of transmitting key material suffer from lack of authentication.

Similarly, a node attack corresponds to taking over the victim's
ability to create certificates. Such an attack is generally harder
than an edge attack, but still feasible. For example, anyone with
physical access to the victim's machine can recover the private key
used to sign certificates, for example by using a keyboard sniffer to
recover the encryption key used to protect the private key.

Edge attack: number of certs. Corresponds to fooling the victim once.

Protection against node attack is a stronger property than protection
against edge attack. Attack of a single node can correspond to an
arbitrary number of edges.

Another factor in trust metric: how many certs are needed? At
simplest, indegree of target node.

We analyze two classes of attacks: one in which the attacker is able
to choose the victims, and another in which the victims are chosen
randomly. The former class of attack is at least as effective as the
latter, and, not surprisingly, we find that it in most cases it is
considerably more so. This result parallels the literature in
scale-free networks, in which removing the ``hubs'' in a scale-free
network with hub-and-spoke topology is far more effective in
fragmenting the network than simply removing random nodes.

\section{Analysis: upper bounds}

Strict upper bounds on effectiveness of trust metric. For node attack,
if $n$ nodes are attacked, entire system falls, where $n$ is minimum
indegree of target node.

Proof: attacker chooses target that is accepted (with minimum
indegree), and chooses its predecessors as victims. Since indegree is
$n$, attack is $n$ nodes. Removing the original target, the graph is
identical to the one in which the original target is accepted, so the
trust metric can't distinguish.

For edge attack, if $n^2$ nodes are attacked, entire system fails.
Choose target as above, then spoof inedges of predecessors of target.

\section{Analysis: lower bounds}

This section analyzes existing trust metrics.

Present simple flow-based metrics, and prove (using min-cut) that they
almost exactly meet the upper bounds given above.

\section{Analysis of Maurer's trust metric}
\label{maurer-analysis}

Analysis of Maurer: follow \cite{LA98}.

\section{Discussion}

Several conclusions follow from the analysis presented in preceding
sections. First, the scalar trust metrics place a very low upper bound
on the attack resistance possible to achieve. Second, we demonstrate
that simple trust metrics based on maximum network flow meet these
upper bounds. Thus, within these assumptions, there is little if any
room to design improved trust metrics.

In particular, analysis of Maurer's trust metric shows its attack
resistance to be both disappointing and equivalent to much simpler
metrics.

A reasonable conclusion, then, is that it is the monotonicity
assumption itself which is flawed, and that it is not possible to
achieve good attack resistance by verifying a small, local subset of
the trust edges comprising the global trust metric. Indeed, in the
next chapter, we show that by relaxing the monotonicity assumption, a
reasonably simple trust metric also based on maximum network flows can
acheive dramatically better attack resistance.
